Author's response to reviews

Title: Streptococcus gordonii septic arthritis : about two cases and review of literature

Authors:

Jean cyr Yombi (jean.yombi@uclouvain.be)
Leila Belkhir (leila.belkhir@uclouvain.be)
Sylvie Jocnkheere (sylvie.jonckheere@uclouvain.be)
Dunja wilmes (dunja.wilmes@uclouvain.be)
Olivier cornu (olivier.cornu@uclouvain.be)
Bernard Vandercam (bernard.vandercam@uclouvain.be)
Hector Rodriguez (hector.rodriguez@uclouvain.be)

Version: 2 Date: 15 December 2011

Author's response to reviews: see over
TO THE EDITOR IN CHIEF OF BMC Infectious Disease Journal

Dear Sir,

Please find enclosed the manuscript entitled “Streptococcus gordonii septic arthritis: about two cases and review of literature”, that we would like to submit for publication in the BMC Infectious disease journal as case report. To the best of our knowledge, Streptococcus gordonii has not yet been described as agent of septic arthritis. We describe here two cases of septic arthritis due to Streptococcus gordonii. It gives us an opportunity to review epidemiology, diagnosis criteria and management of septic arthritis.

This manuscript has been reviewed and has been approved by all authors. We accept the responsibility for the content. We confirm that the manuscript has not been submitted or is not simultaneously being submitted elsewhere and that there is no conflict of interest.

We thank you for your editorial work.

Consent

Written informed consent was obtained from the two patients for publication of this case report and any accompanying images. A copy of the two written consent are available for review by the editor-in-chief of this journal.

Sincerely yours.
Jean Cyr YOMBI

Department of Internal Medicine and Infectious Diseases
Cliniques Universitaires St Luc, UCL
10, Avenue Hippocrate, 1200 Brussels, Belgium
Tel: +32 27643107
Fax: +32 27649058
Email: jean.yombi@uclouvain.be